Author’s response to reviews

Title: Endocrine Manifestations in POEMS Syndrome: A Case Report and Literature Review

Authors:

LI hui (zehui2006@126.com)
Yumeng Huang (huangyumengnc@126.com)
Yan Li (hawthornli@163.com)
Baozhong Zheng (baozhongzheng@sohu.com)
Ming Liu (mingli@med.umich.edu)

Version: 2 Date: 29 Nov 2017

Author’s response to reviews:

Dear editor :

We have revised our case report due to your criterion, The Declarations was written follow the Conclusions section, and before the References and include parts as follows:

Abbreviations
Ethics approval and consent to participate
Consent for publication
Availability of data and material
Competing interests
Funding
Authors' contributions
Acknowledgements

Please check it and if any should be changed, please note me.

Thank you very much for your patience!

LiuMing